# The Effect of Banking Relationship on Firm Performance in Malaysian Public Listed Companies

Putri Aidatul Sofina Binti Mohamad Jamil

College of Business

Universiti Utara Malaysia

July 2015

# The Effect of Banking Relationship on Firm Performance in Malaysian Public

Listed Companies

By

# Putri Aidatul Sofina Binti Mohamad Jamil

814436

A Thesis Submitted to the Fulfillment of the Requirement for the Degree Master of Science (Management)

**College Of Business** 

Universiti Utara Malaysia

July 2015

@ Putri Aidatul Sofina Binti Mohamad Jamil, 2015. All Right Reserved

### DECLARATION

I am responsible for the accuracy of all opinion, technical comments and illustrations in this thesis except for the citations and quotations that have been adequately acknowledged. I bear full responsibility for the checking whether material has been previously or concurrently submitted for any other master's program at UUM or other universities. UUM does not accept any liability for the accuracy of such comment, report and other technical information claims.

#### PUTRI AIDATUL SOFINA BINTI MOHAMAD JAMIL, 814436

College of Business

Univesiti Utara Malaysia

06010 Sintok Kedah, Malaysia

July 2015

#### **PERMISSION TO USE**

In presenting this thesis in partial fulfillment for a post graduate degree from the Universiti Utara Malaysia, I agree that University Library may make it freely available for inspection. I further agree that permission for copying of this thesis in any manner, in whole or in part, for scholarly purposes, may be granted by my supervisor or, in her absence, by the Dean of Research and Postgraduate Studies, College of Business. Any financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and to Universiti Utara Malaysia for any scholarly use which may be made of any material from my thesis.

Request for permission to copy or to make other use of materials in this thesis, in whole or in part should be address to:

#### **Dean of Research and Postgraduate Studies**

**College of Business** 

Universiti Utara Malaysia

UUM, 06010 Sintok, Kedah, Malaysia

#### ABSTRACT

Funds from banks play an important role in the growth and maintenance of a firm. Hence, it is important that a firm creates and maintains a good relationship with its bank in order to secure the funds. Establishing a good bank-firm relationship can also help to reduce conflicts between shareholders and creditors since close bank relationship helps the firm to get funding from other financial institutions. Relationships between a bank and a firm consist of two types: bank-borrower relationship and consumer-supplier relationship. The bank-borrower relationship involves primarily a loan agreement between two interested parties. Cooperation between a bank and a firm is more durable than customer-supplier relationships in terms of provision of financing. This thesis investigates the effect of banking relationships on firm performance on companies in Malaysian capital market. Seventy eight firms from the Top 100 public-listed companies in Bursa Malaysia are examined. Investigation is performed to show the effects of number of banking relationships, short-term financing, long-term financing, firm size, and foreign ownership on the firms' performance as measured by return on assets (ROA), return on equity (ROE), and Tobin's Q. Findings show a significant and negative relationships on the number of bank relationships for all of firm performance measures, consistent with previous studies. For ROA measurement, result shows a significant positive relationship with short-term financing, a finding also consistent with past studies. Short-term loan and long-term loan impact ROA and ROE positively, while affecting negatively firm performance measured by Tobin's Q. ROA and ROE have a positive correlation with foreign ownership, while Tobin's Q is negatively correlated with foreign ownership. In general, the study contributes to banking literature by investigating and evaluating the relationship of firm performance and lending banks in shareholders' perspectives.

#### ACKNOWLEDGEMENT

All praise to Allah S.W.T., the Lord of Universe. Blessing and peace be upon our prophet Muhammad S.A.W for giving guidance, courage and spirit to us in order to complete this thesis.

This thesis was written during the fourth semester of 2014/2015 at Universiti Utara Malaysia in fulfillment of the Master Program in Science Management. Many people have helped me with the completion of this study and make the work possible.

I would like to express my sincere gratitude to my supervisor Associate Professor Norafifah Ahmad for her advice, guidance, support, patient, knowledge and encouragement throughout the completion of my thesis. I extend my special gratitude to all my post-graduate lecturers, and my great appreciation for their exciting classes. I also would like to acknowledge the kind assistance of staff from the library and main Office of COB for their help.

I would like to express me special thanks to my family for their support and encouragement throughout my life, my friends, as well as my classmates, not forgetting those who have contributed to my life especially my family and my friends.

Thank you for your support. Only Allah can repay all your help and kindness.

Putri Aidatul Sofina Binti Mohamad Jamil College of Business Universiti Utara Malaysia

# TABLE OF CONTENT

	Page
DECLARATION	Ι
PERMISSION TO USE	II
ABSTRACT	III
ACKNOWLEDGEMENT	IV
TABLE OF CONTENTS	V
LIST OF TABLES AND FIGURES	VIII
LIST OF ABBREVIATIONS	IX

# CHAPTER ONE: BACKGROUND OF STUDY

1.1	Introduction to Study	1
1.2	Problem Statement	5
1.3	Bank-Firm Relationship	6
1.4	Research Questions	9
1.5	Research Objectives	10
1.6	Significance of the Study	11
1.7	Organization of Remaining Chapter	11

## **CHAPTER TWO: LITERATURE REVIEW**

2.1	Introduction to the Study	13
2.2	Bank-firm Relationship	13
2.3	Firms' Performance	14
2.4	Banking Relationship	15
2.5	Credit Financing Relation	18
2.6	Firm Size	21

Foreign Ownership	21
Lending Relationship	22
2.8.1 Lending Relation Value	25
A. Information from Relation	25
2.8.2 The Empirical Value of Lending Relations	27
Advantages of Lending Relationship	33
2.9.1 Exclusive Bank Relationship	33
2.9.2 Multiple Bank Relationship	35
Disadvantages of Lending Relationship	37
2.10.1 Exclusive Bank Relationship	39
2.10.2 Multiple Bank Relationship	40
	Foreign Ownership Lending Relationship 2.8.1 Lending Relation Value A. Information from Relation 2.8.2 The Empirical Value of Lending Relations Advantages of Lending Relationship 2.9.1 Exclusive Bank Relationship 2.9.2 Multiple Bank Relationship Disadvantages of Lending Relationship 2.10.1 Exclusive Bank Relationship

# **CHAPTER THREE: METHODOLOGY**

3.1	Introduction		42
3.2	Theoretical Framework		42
3.3	Hypothesis Development		43
3.4	Data C	Collection	44
	3.4.1 \$	Sample Selection	44
	3.4.2 0	Ordinary Least Square Regression and Model Specification	45
		3.4.2.1 Model Specification	46
3.5	Resear	ch Model and Measurement of Variables	47
	3.5.1	Dependent Variables	47
	3.5.2	Independent Variables	48

# **CHAPTER FOUR: RESULTS AND FINDINGS**

4.1	Introduction	52
4.2	Descriptive Analysis	52
	4.2.1 Sector Classification	53
	4.2.2 Number of Bank Relationships	53

	4.2.3 Percentage of Foreign Ownership of Sample Companies	54
4.3	Descriptive Statistics	55
4.4	Multicollinearity	57
4.5	Correlation Analysis	58
4.6	Regression Results	60
4.7	Discussion	64

## **CHAPTER FIVE: SUMMARY AND CONCLUSION**

5.1	Introduction	67
5.2	Summary of General Findings	67
5.3	Limitations of the Study	69
5.4	Recommendations for Firms	70
5.5	Recommendations for Future Research	72
5.6	Conclusion	73

REFERENCES	74
APPENDIX	82

# LIST OF TABLES, FIGURES AND GRAPHS

Table 3.1	Independent and Control Variables	50
Table 3.2	Description of Variables Used in the Study	51
Table 4.1	Sample Distribution Based on Sector Classification	53
Table 4.2	Percentage of Bank Number Relationship of Sample	54
	Companies	
Table 4.3	Percentage of Foreign Ownership of Sample Companies	55
Table 4.4	Descriptive Statistic of Continuous Variables	56
Table 4.5	Multicollinearity Test Summary	58
Table 4.6	Correlation Matrix Summary	59
Table 4.7	Linear Regression Model Summary	60
Table 4.8(a)	Linear Regression Analysis (ROA)	61
Table 4.8(b)	Linear Regression Analysis (ROE)	62
Table 4.8(c)	Linear Regression Analysis (Tobin's Q)	63
Table 5.2	Acceptability of the Hypothesis	68

43

# LIST OF ABBREVIATIONS

BNR	Number of bank relations
FBMKLCI	FTSE Bursa Malaysia Kuala Lumpur Composite Index
FSZ	Firm Size
FOW	Foreign ownership
LTF	Long-term financing
LAsset	Log of asset
ROA	Return on asset
ROE	Return on equity
STF	Short-term financing

# Chapter 1 Introduction

#### **1.1 Introduction to Study**

As the business world today becomes more competitive and challenging, firms must have good management in order to improve their performance and sustain it. The management includes the board of directors as a resource provider because they provide social and business contacts, and influence the environment in favors of the firms (Carpenter and Westphal, 2001). It is also believed that managers have the capability as a resource to provide connections to other business sectors and also to the external environment (Zahra and Pearce, 1989). However, there is a separation between management and owners, where the managers tend to act with self-interest that are not always in the best interest of the owners (Jensen and Meckling, 1976).

In order to avoid any irresponsible and unethical acts, the regulatory bodies and shareholders are required to impose a monitoring and control process via corporate governance code to control risks. Corporate governance generally refers to the mechanisms, processes and relations by which firms are controlled and directed. It identifies the rights and responsibilities among different participants in the corporation (such as the board of directors, managers, shareholders, creditors, auditors, regulators, and other stakeholders) and includes the rules and procedures for making decisions in corporate affairs. Corporate governance includes the processes through which corporations' objectives are set and pursued in the context of the social, regulatory and market environment.

Bank-firm relationships are linked with the main functions of banks which are keeping the customer's account, and implementing loan agreements or other banking services. They involve exchange of benefits and information over time between the bank and the enterprise, leading to development of certain information resources, both on the bank's part and on the firm's part. It is assumed that the bank and the firm are in a relational contract if they have an understanding which allows certain contract terms and conditions to be further specified over time. Over a longer time period, the firm relies on the bank, which provides it with financial services, whereas the bank depends on repayment of loans by its long-term borrowers and the borrowers purchasing loan-related services (Heffernan, 2007).

There are two types of relationships between a bank and a firm: bank-borrower relationship and consumer-supplier relationship. The bank-borrower relationship involves primarily a loan agreement between two interested parties. The most crucial research problems undertaken in the field of relationship banking include determining its characteristics and evolution under the influence of changes in markets for banking services, changes in the structure of financial systems (including technological changes) as well as its impact on terms and conditions of financing firms, goodwill of firms and banks. Cooperation between a bank and a firm is more durable than customer-supplier relationships in terms of provision of financing. Hodgman (1961) demonstrates that firms with bank deposits enjoy better lending terms and conditions than firms without deposits. However, according to Wood (1975), the so-called lending relationships may develop independently and banks may extend loans at a lower price, hoping for higher income in the future. During long-term provision of financial services, the bank collects additional information about its customer. Information about the borrower is an important aspect of banks' lending decisions, thus its quality should improve proportionally to the duration and closeness of the cooperation (Diamond 1984; Petersen and Rajan, 1994).

In order to conduct flexible firm financing policy, the bank should be in a close relationship with the firm. This relationship is referred to as relationship banking. Banks benefit from maintaining such relationships through better access to information about the firm, whereas the firm expects the bank to provide access to financing even when it experiencing financial difficulties (such as during financial crisis). Boot (2000) defines relationship banking as provision of financial services by a bank which: (1) invests in access to specific information about a customer, which is frequently publicly unavailable and (2) assesses the profitability of this investment while taking into account the duration of cooperation with the customer and banking products provided to it.

Berger and Udell (1998) define relationship banking (relationship lending) as the bank being able to obtain information about the firm during cooperation. The information is then used to develop future business conditions of bilateral cooperation (loan availability, interest rates and collaterals). To sum up, relationship banking requires four conditions to be fulfilled: (1) banks being in contractual relationships with enterprises, (2) as a result of relationships banks enduring over time (3) collecting information more extensive than that available from public resources and (4) the collected information being confidential (Berger, 1998).

However, it should be noted that the nature of cooperation between a firm and a bank is more complex and it is difficult to provide strict definitions of relationship banking and transaction-oriented banking. To simplify it, one can assume that if a firm uses services of only one bank at a given time (so-called single relationship), this cooperation takes the form of relationship banking. If there are more banks providing credit facilities, we are dealing with multiple relationships or transaction-oriented cooperation. However, it should be noted that multiple relationships may also have the nature of relationship banking. As a result, banks may reduce the monopoly margin and offer better financing terms to current customers (Petersen and Rajan, 1994). This cooperation usually involves a so-called main bank.

Banking relationships on firm performance have been widely studied in developed countries. However, to the present author's knowledge, this topic has not been much examined empirically in Malaysia. Hence, the purpose of this paper is to analyze whether firm's banking relationship affects the performance of firms in Malaysia and to suggest policy implications related to the firm's banking relationship.

This study examines the performance of firms in the bank-firm relationship on selected public listed companies in Malaysia. The analysis is based on sample from top 100 public listed firms in Malaysia (excluding financial firms) in the year of 2013. This provides a final

sample of 78 non-financial firms. In particular, this study attempts to determine whether firms with strong banking relationships operate more effectively compared to the other firms that do not enjoy banking relationship. Since research in this area is still lacking in Malaysia, this current study attempts to provide an analysis on firm's banking relationship and show how this relationship affect firms' performance.

#### **1.2 Problem Statement**

Previous studies regarding bank-firm relationship carried out in other countries have focused more on the firms' side. It is believed that establishing a good bank-firm relationship will help to reduce the conflicts between shareholders and creditors. Some authors argue that close bank relationship helps the firms get funding from other financial institutions (Diamond, 1984; Fama, 1985; Horiuchi, Parker and Fukuda, 1988).

One of the corporate governance mechanisms is through board of directors, which are generally believed to have an important role in corporate governance, especially in monitoring top management (Fama and Jensen, 1983). The boards have the responsibility to monitor the management in order to protect the shareholders and at the same time to ensure that the managements acts in shareholder's interest.

In Malaysia, there are many firms listed on the stock market exchange with the purpose to help the firm gaining other funding resources. However, funds from the bank still play an important role in growth and maintenance of the firms. Hence, it is important to the firms to have good relationship with the bank in order to secure the funds. Despite the positive impacts of the bank-firm relationship, there are also arguments about the effectiveness of the firm with better banking relationship with those firms which do not have such relationship.

This present study contributes to the literature on banking relationship by examining and establishing whether bank-firm relationship has a positive or negative impact on firms' performance. The dependent variable of this study is firm performance, which is measured by two measurements from different perspectives: accounting measures (ROA and ROE) and a market measure (Tobin's Q). The independent variables are the number of bank-firm relationships, and bank credit financing (short-term, long-term, and overall). Two control variables (firm size and foreign ownership) are also included.

#### **1.3 Bank-firm Relationship**

Relationship banking is more frequently encountered in countries such as Japan and Germany, where there are strong capital relationships between banks and enterprises from the non-financial sector (Ongena and Smith, 2000). In the previous century in Japan and Germany, close bank-firm relationships were considered to be one of the key success factors for these economies. However, in the 1990s, the popularity of relationship banking dropped due to firms having more opportunities of obtaining funds and increased number of players in the financial market (Haffernan, 2007).

Bank-firm relationships depend on several factors. The benefits and costs related with the changes in these relationships depend on business cycles and the firm which provide incentives to interested parties to establish and support such relationships (Degryse and Ongena, 2001). These relationships have impact on both parties respectively. From the banks' perspective, relationship financing reduces negative selection linked with information asymmetry, which is particularly increasing during periods of financial crises or restrictive monetary policy. On a firm's point of view, relationship financing increases access to external financing.

Long-term relationship between main banks and their client firms generate value and increase economic efficiency. In the process of building the relationship, the lender gains borrowerspecific information which gives him several benefits. These relationships will also be valuable from the borrower's point of view to the extent that the lender passes these benefits to the borrower.

Even though commercial banks cannot directly trade in shares, they may still manage the portfolios held in trust on behalf of customers (Santos and Wilson, 2005). Moreover, commercial banks are often part of the bigger financial conglomerates, with affiliated investment associate (such as investment banks, mutual funds, pension funds, and insurance companies) that can trade on the basis of the information acquired through lending (Massa and Rehman, 2008; Acharya and Johnson, 2007). Thus, the privileged information of the commercial bank and its potential to influence the borrower's stock price by trading through its asset-management department may increase information asymmetry and adverse selection

for the investors in the borrower's stock. This creates disincentives for other investors to trade in this stock, thus lowering its liquidity.

In the context of asymmetric information in credit markets, lending relationships make the information exchange between the borrower and the lender easier through repeated interaction over the duration of the relationship and through the provision of multiple financial services. Lenders invest in generating information from their client firms and borrowers are easier to disclose information (Boot, 2000).

It is widely accepted that bank loans are special because of inside information that comes with lending. More inside information increase the monitoring ability of the bank (Diamond, 1984), and therefore improves capital allocation and corporate governance (Levine, 2002). However, the bank may also exploit its informational advantage in the equity market and effectively become an insider (Kahn and Winton, 1998). This dual effect on the borrower makes bank loans special.

However, the information asymmetries between the bank and the firm are lessened as time goes by. This process enhances economic efficiency through many channels. Firstly, having a long-term horizon facilitates the design of implicit credit contracts over the duration of the relationships that may increase value. Second, the reusability of the information generated by the lender over repeated transactions and over time is also beneficial in terms of savings on the fixed cost of screening and monitoring (Boot, Greenbaum and Thakor, 1993). Third, it avoids the free-rider problem of monitoring since the bank internalizes the benefits of such investments. Higher monitoring levels increase value since, for instance, they help solve principal-agent problems of managerial behavior.

Close bank-firm relationships have some disadvantages to the firm. The most significant disadvantage is that having a single relationship gives an informational monopoly to the only informed bank, which can impose hold-up costs for the firm (Sharpe, 1990; Rajan, 1992). Additionally inefficient loan renewal decisions is more likely to happen when only one lender has the option to bail out the firm in case of financial crisis (Dewatripont and Maskin, 1995) where managers are more disposed to divert cash to themselves when there is only one creditor than when there are many creditors (Bolton and Freixas, 2000). However a firm will benefit from relationship lending as long as the bank shares the value with the borrower. Consequently, if lending relationships are valuable, it should be reflected in the overall firm performance.

#### **1.4 Research Questions**

In order to conduct the analysis of the study, the following research questions are posed:

- 1. Is there a significant relationship between the numbers of bank relationships and the firms' performance?
- 2. Is there a significant relationship between short-term financing and the firms' performance?

- 3. Is there a significant relationship between long-term financing and the firms' performance?
- 4. Is there a significant relationship between firm size and firm performance?
- 5. Is there a significant relationship between foreign ownership and firm performance?

# **1.5** Research objectives

The main purpose of this research is to examine firm performance in the bank-firm relationship in Malaysian firms. Do firms that have better banking relationship actually operate more effectively compared to those which do not have? In addition, the study has three sub-objectives as follow:

- 1. To examine if there is a significant relationship between the number of bank relationships and firm performance.
- 2. To examine if there is a significant relationship between short-term financing and firm performance.
- 3. To examine if there is a significant relationship between long-term financing and firm performance.
- 4. To investigate whether there is a significant relationship between firm size and firm performance.

5. To establish if there is a significant relationship between foreign ownership and firm performance.

#### **1.6** Significance of the study

The significance of this thesis involves two important aspects, theoretical and practical. The study describes the theoretical background which forms the basis for empirical prediction and hypotheses. Since there are not many studies examining this bank-firm relationship in Malaysia, the results are expected to contribute new knowledge in this area. The samples from the top 100 public listed companies have a big impact on the FBMKLCI and since foreign and local investors usually take into consideration the performance of the Top 100 public listed companies in order to invest, the results therefore should be able to provide an influence on shareholders' and investors' decision makings.

# 1.7 Organization of Remaining Chapter

The rest of this paper is organized as follows. Chapter One provides an overview of the study and explains the important of the study including the problem statement of the study, significance and also objectives of the study. Chapter Two focuses on bank-firm relationships, which provide a review of relevant literature on bank-firm relationship as the basis of this research, and firm's performance. Chapter Three provides the methodology adopted in this study, simultaneous research design; the sample, data collection, research's model and measurement variables. Chapter Four discusses the results of the study using tables and observations. Chapter Five concludes the thesis.

# Chapter 2 Literature Review

#### 2.1 Introduction to Study

This chapter explores and discusses relevant literature reviews which focus on many aspects of bank-firm relationship. The discussion is divided into three sections. First section presents the theoretical basis of bank-firm relationship. Second section discusses about the positive impact of the bank-firm relationship to the firm performance, and the third section will represent the negative impact of the bank-firm relationship to the firms' performance.

## 2.2 Bank-firm Relationship

A bank relationship is a continuing contact between the financial institution and the firm for the provision of financial services beyond simple, various transactions which is associated with the collection of information that can be used to make decisions about the evolution of the contract terms (Berger and Udell, 1998). On the other hand, borrowing firms may need to consider negative effects from the leakage of confidential information. The main issue that arises is the potential benefit of this relationship for the borrower and the bank. This benefit has to be taking into consideration the influence of external factors such as the competitiveness of the environment, the degree of technological diffusion and the level of financial market development. The banking relationship consists of two type of relation: deposit relation and lending relation. This study will concentrates on the lending relationships. Banking relationships are shown through the number of banking relationships, duration of banking relationship, the amount of credit (Peltoniemi, 2004), interest (Bolton and Freixas, 2000), and banking services (Degryse and Cayseele, 2000).

Banks benefit from maintaining such relationships through better access to information about the firms, whereas the firm expects the bank to provide access to financing even when it experiencing financial difficulties. According to Ongena and Smith (2000), relationship banking can be described with more detail using two dimensions. The first dimension is duration since the measure of relationship banking is the length of bank-firm cooperation (Wood, 1975; Rajan, 1997). The second dimension is the product scope of cooperation (Hodgman, 1963).

#### 2.3 Firms' Performance

Castell, Dwyer and Hasan (2006) shows that ROA and ROE of Italian enterprises were dropping in 1998–2000 as the number of lending bank increased (multiple-banks), especially among small firms. The results of these works support the thesis that a single-bank relationship reduces information asymmetry and agency costs and allows to neutralize the effect associated with the so-called monopoly rent (hold-up problem).

Bank-influenced firms should have better performance than independent firms, if banks have access to private information that is used to reduce agency costs, and if firms with close bank relationships benefit from better access to finance. These relationships should lead to both higher profitability and higher growth rates for bank-influenced firms (Argawal and Elston, 2000).

Firm performance can be negatively affected by bank-influence if banks choose to fully use their unique position for their own interests rather than the shareholders' interest. For example, banks could share private information about the firm with its competitors or strategically release industry specific information to better its own interests at the expense of the firm. Additional conflicts of interest could also include the bank influencing management to undertake less risky projects in case that the bank is financing the project.

### 2.4 Banking Relationship

Exclusive bank-firm ties not only affect the price and availability of capital, but also may also directly influence management decisions and firm performance. Maintaining multiple creditors creates inefficient renegotiation which may cause strategic default (Bolton and Scharfstein, 1996). Banks may want to diversify firm-specific credit risk, resulting in more creditors for high default risk firms. The extant empirical evidence, though for smaller and medium sized firms, seems to indicate such diversification may play a role. Distressed firms (in Germany and Italy respectively) have significantly more bank relationship compared to non-distressed firms (Harhoff and Körting, 1998; Foglia, Laviola and Marullo, 1998).

Firms are often required to disclose proprietary information to demonstrate their quality in order to obtain credit. Checking account activity may lead to an additional source of 'confidential' information about a firm's current condition and future prospects (Vale, 1993; Nakamura, 1993). A bank could transfer such valuable information, directly or indirectly, to product market competitors of the firm once the information acquired and analyzed.

However, a bank might stop spreading confidential information obtained in an exclusive relationship when there are losses of reputation for the bank, or the negative impact on the performance of an important client may. Banks may still accidentally, leak confidential information. The information spill from the bank may be limited or they may choose to provide useful information to an important client. The provision of information and advice to firms has also been an important part of the bank-firm relationship (Hellwig, 1991).

Recent researches focus more closely on the interaction between the number of creditors and product market competition, with creditors acting as intermediaries of confidential information. Based on Yosha (1995), when an innovator is entering a competitive product market with one established competitor, there is asymmetric information about the impact of the innovator's entry on the profits of the established firm. If the entrant draws a high quality project, the profits of the established firm will be decrease. These expected losses will trigger an aggressive reaction by the established firm to reduce the expected profits of the innovator. As a result, the high-quality innovator would like to hide its product as long as possible. However, if the innovator draws only low-quality projects, the innovator would like to reveal

their product as soon as possible to avoid any unnecessary aggressive response from the established firm.

The number of financing sources determines the revelation percentage of the innovator's confidential information concerning project quality (Degryse and Ongena, 2000). If the innovator obtains financing from one source (exclusive financing), less information is leaked to the established competitor than if the innovator uses multiple financing. However, a multilateral financing arrangement is more costly because it is more difficult to communicate with multiple lenders, and there may be a loss for the borrowing firm, as its actions have to be coordinated with more than one lender.

Multilateral banking does not only cause higher transaction costs, but also more competitive interest rates and the innovator can decide whether or not to disclose confidential information during the loan granting process (von Rheinbaben and Ruckes, 1998). In other words, establishing a bank relationship and disclosing confidential information are independent decisions. If the innovator discloses information, the innovator's expected sales profitability will be reduce by the lender banks, as the probability that information leaks to the established firms increases. If there is no disclosure of confidential information, sales profitability remains unaffected.

As a result, there is no continuous relationship between sales profitability and the number of banks. Sales profitability decreases in the number of banks within the range of disclosure, and remains high when no disclosure takes place. Only a highly-rated innovator will find it

optimal not to disclose confidential information, while dealing with multiple banks. Disclosure would not improve the innovator's credit standing, and without disclosure there is no risk of information leakage, while competition between multiple banks lowers the interest rate.

Summarizing, multiple bank relationships go hand-in-hand with lower sales profitability in study by Yosha (1995) and von Rheinbaben and Ruckes (1998) in case of information disclosure. A highly rated innovator will not disclose confidential information such that its high sales profitability is unaffected. This innovator will deal with many banks in order to obtain a competitive interest rate.

#### 2.5 Credit Financing Relation

The literature discussed so far focus on the static relation between the number of credit relationships and the cost and availability of capital. There are additionally connections between the credit relationships switching, firm quality, and the cost of capital (von Thadden, 1998). The informational asymmetry between the inside bank and the outside banks gives the existing lender an advantage. The inside bank observes after one period the quality of the firm (good or bad), and can therefore set the interest rate in the second period contingent on the quality of the firm. The other 'outside' banks observe only from the rumors about the firm's quality.

In particular, the lower the interest rate offered, the higher is not only the probability that the bank obtains the client-firm, but also the higher is the probability that the firm, if obtained, is estimated by other banks to be of a lower value. By switching banks, both good and bad firms reduce their interest payments, compared to non-switching good and bad firms respectively. But bad firms are expected to switch more frequently.

Jensen and Meckling (1976) argue that possibility of problems between the managers, shareholders and creditors tend to increase the firms' external costs of finance. Bank-influenced firms should enjoy increased access to capital through easier access to bank debt or preferential terms on loans. In addition, bank involvement with a firm serves as a signal to outside investors and causes a certification effect, which makes it easier for firms to gain funds.

Close bank-firm relationships can also cause a potential conflict of interest that may show in firm financing decisions. Specific abuses of private information held by banks may include influencing the firm to issue equity to pay bank-debt in cases of financial issue or the use of equity rather than debt financing to finance risky projects. Thus, leverage could either be higher or lower for bank-influenced firms, but close relationships with the bank should improve the firm's access to bank debt.

Part of the literature on the value of a close relationship between a firm and a bank is premised on observations by Fama (1985) and James (1987) compared to other forms of

financing, bank loans to firms are based on less widely available information. The relationship between a firm and a bank can help to overcome information asymmetries and agency problems that create liquidity constraints which can reduce firms' investment (Fazzari, Hubbard and Peterson, 1988; Hoshi, Kashyap and Scharfstein, 1991; Bernanke and Gertler, 1995). These theories suggest that screening and monitoring by banks can overcome information and possibility of problems and reduce liquidity constraints for borrowers (Leland and Pyle, 1977; Diamond, 1984; Boyd and Prescott, 1986; Bhattacharya and Thakor, 1993).

Bank-firm relationships may affect firm profitability through the price of capital. Multiple bank relationships decrease the informational lock-in problem, which may reduce the interest rate charged by the inside banks (von Thadden, 1992). Some evidence supports this potential correspondence between the number of bank relationships and the interest rate paid by the firm. The numbers of credit relationships show a negative effect of on the interest rate charged by the credit granting banks (Angelini, Di Salvo and Ferri, 1998).

However, Petersen and Rajan (1994) find that, for small U.S. firms, multiple bank relationships may even increase lending rates. Degryse and Van Cayseele (2000) show that Belgian firms which took loans from a second bank pay higher rates on these loans compare to the interest rates these firms pay on the loans granted by their first bank.

#### 2.6 Firm Size

A study done by Thanh and Minh Ha (2013) shows that firm's size is related positively to firm's performance because small firms are less likely to be close to the minimum efficient scale needed to operate efficiently in a market (Audretsch and Mahmood, 1994; and Geroski and Mata, 2005) and small firms also apply different technologies in accordance with their size. Due to their small scale, they use less capital intensive approaches. Because operational risk is inversely related to company size, small companies should fairly have less debt. In a case of lowering prices, small firms find it difficult to bear these costs for a long time; thus, they are less efficient.

Furthermore, small firms' market power is weak, and their operations are in niche markets, which are uncertain. They are more vulnerable to temporary demand shocks and the uncertainties of the market than large firms. Moreover, small firms are typically less diversified than larger firms, which may deteriorate their survival prospects by increasing risk and failing to keep alive options in a fiercely competitive market (Geroski and Mata, 2005).

There are a number of ways to measure size of a firm, for example size of company measured based on natural logarithms of sales (LNSA) and logarithm of book value of firm assets.

#### 2.7 Foreign Ownership

Wei, Xie and Zhang (2005) and Tian and Estrin (2008) suggest a positive relationship between firm performance and foreign shareholdings. They suggest that foreign investors can effectively monitor and influence management of the firm that they are holding. If foreign ownership represents only 0% to 5%, it is only a small shareholder and it has less power of voting. If the foreigner owns from 5% to 35% firm's equity, it is a major shareholder, but it is not strong enough to vote on any business issues of firm. However, if the foreigner holds shares of 35% or more, it is a major shareholder and has enough power of vetoing on any business issues of firm.

#### 2.8 Lending Relationship

There is a consensus that banks involved in relationship lending also offer other services which do not require such close relationships or are not binding for the parties in a long-term perspective. Such cooperation is referred to as transaction-oriented banking. In transactionoriented banking, the parties develop agreements so as to choose the most beneficial terms and conditions.

However, it should be noted that the nature of cooperation between firms and a bank is more complex and it is difficult to provide strict definitions of relationship banking and transaction-oriented banking. If the firm uses services of only one bank at a given time which known as single relationship, this cooperation takes the form of relationship banking. Otherwise, when there are more banks providing credit facilities, multiple relationships or transaction-oriented cooperation exist. However, it should be noted that multiple relationships may also have the nature of relationship banking. As a result, banks may reduce the monopoly margin and offer better financing terms to current customers (Petersen and Rajan, 1994). This cooperation usually involves a so-called main bank. Relationship banking is more frequently encountered in countries such as Japan and Germany, where there are strong capital relationships between banks and enterprises from the non-financial sector (Ongena and Smith, 2000).

A main objective of financial markets is to transfer capital from investors to firms and entrepreneurs with the intention to earn profit from the investment opportunities and divide the risk among investors. The existence of financial markets and financial institutions occurs when they arise to help solve some of the issues that can be found in real markets.

Banks offer multiple purposes. They transform short term cash investments such as deposits into long term assets such as loans (Diamond and Rajan, 1998). They also focus on collecting and processing the information necessary to make investment and lending decisions. If borrowers know more about their abilities and intentions, capital markets can break down because there may be no price for capital (interest rate) which will allow lenders to make a profit (Stiglitz and Weiss, 1981). This occurs when the information about borrowers is limited and expensive to acquire.

Banks are more efficient at collecting information due to simple economies of scale. They can collect information of hundreds of borrowers at once thus reducing the cost of collecting information. Theory suggests that a firm with close relationship with financial institutions should have a lower cost of capital and greater availability of funds compare to a firm without such relationship, if the information collected can be used as an input to the lending decision over multiple periods and not easily duplicate by competitors (Diamond, 1984, Haubrich, 1989; Diamond, 1991).

The idea that banks can provide a service in the form of a lending relationship arises out of the value that firms place on these relationships. If by building a relationship banks can learn the quality of borrowers, they may be able to profitability lend, where less informed capital providers may not.

Research shows that building a relationship with a lender has only a small effect on the cost of capital facing the lending firms. Instead, relationships appear to be valuable because they give these firms access to more capital, not because they provide small firms with cheaper capital. The role relationships play in providing capital to small firms depends mostly on the competitiveness of the capital markets. The number of bank relationships can increase with a firm's age and size (von Rheinbaben and Ruckers, 2004).

A difficult situation in the banking sector works to the advantage of relationship banking, which has been confirmed by studies related to crises in Asia. In conditions of a crisis, banks maintaining close relationships with firms may subsidies their customers by offering better crediting conditions, in particular for small, loyal firms (Berg and Schrader 2009). Empirical results have demonstrated that strong, durable relationships with banks had positive impact on development of firms and their fate during economic crises (Fok, Chang and Lee, 2004).

The bank-borrower relationship, which involves mainly a loan agreement between two parties, remains the subject of research related to its nature and effects — for banks, firms as well as development of financial systems (Koch and Macdonald, 2000). Cooperation between a bank and a firm, focused on provision of financing, is more durable than other customer-supplier relationships. Bank-firm relationships are associated with the main functions of banks including keeping the customer's account, implementation of loan agreements or other banking services. They involve exchange of benefits and information over time between the bank and the enterprise, leading to development of certain information resources, both on the bank's part and on the firm's part.

#### 2.8.1 Lending Relation Value

#### A. Information from Relation

Theoretical analyses suggest that a close relationship between a bank and a firm can reduce information asymmetries, improve the firm's access to credit and lead to an overall improvement in the firm's performance. Stiglitz and Weiss (1981) find that the threat of future credit rationing can reduce moral hazard. Diamond (1991) shows that reputation building through bank borrowing can provide certification, which can allow a firm to eventually raise funds on public markets. This benefit of a successful bank relationship raises the cost of default on a bank loan and lowers the probability of default.

In a simple market, price will adjust according to supply and demand for a good. Thus in the loan (capital) market, the interest rate should adjust according to the demand for capital with the supply of capital. The interest rate would rise until the excess demand for capital was
driven to zero if firms found themselves wanting more capital than lenders were willing to supply at the current price.

Firms with higher risk would need to pay higher rates on their capital. This simple rules works as long as lenders know information about the borrower's type and investment opportunities same as the borrower. Otherwise increases in the interest rate may not clear the market. Higher interest rates will cause safer borrowers to choose other banks or encourage firms which accept the loans to invest in riskier projects. Thus the higher interest rate may no longer cover the bank's expected losses and the bank may optimally choose not to lend (Stiglitz and Weiss, 1981). This causes the firms to faced credit constrained especially when having more profitable investment than they are able to fund.

One solution to this problem is the use of financial intermediaries to produce and use information in the loan origination process. The market can once again clear if the information between borrower and lender can be resolved. The role of credit reporting agencies in transforming information from multiple sources into easy to use indexes is part of this process. However, not all information can be collected from information agencies.

During long-term provision of financial services, the bank collects additional information about its customer. Information about the borrower is an important aspect of banks' lending decisions and its quality should improve based on the duration and closeness of the firms (Diamond, 1984; Petersen and Rajan, 1994). In order to conduct flexible firm financing policy, the bank should be in a close relationship with it. This relationship is referred to as relationship banking. In the abovementioned context, Ongena and Smith (2000) define relationship banking as a relationship between a bank and a firm, which is something more than a simple, anonymous financial transaction

#### 2.8.2 The Empirical Value of Lending Relations.

Lending relationships should be most valuable where the information about a firm and its potential investment opportunities are most uncertain. Empirical research on lending relationships focused on small firms due to they tend to be young and have little track record. They are often in new industries or markets, so firms which they can be compared are also less common.

However, for large and publicly traded firms, lending relationship appears to have value due to the access to capital markets are less costly and lending relationship potentially less valuable. Slovin, Sushka, and Polonchek (1993) examine the stock price response of firms which had publicly disclosed lending relationships with Continental Illinois Bank when the bank announced its insolvency. The stock price change was bigger when the borrower had publicly documented relationships with other banks.

# 2.9 Advantages of Lending Relationship

The most important advantage of relationship banking is it minimized agency costs on the principal-agent line (which occurs as a result of a contract between the lending bank and the firm) due to the bank gains additional information and decreases the costs resulting from the

issue of negative selection (Heffernan, 2007). For firms to maintain multiple relationships is expensive, mainly due to transactional costs (Diamond, 1984). However, empirical research provides no explicit answers, while costs and benefits depend on many factors.

Relationship financing leads to two-way information flows, which the parties getting to know their mutual expectations, deeper recognition of the customers' situation as the basis for trust and improved flexibility of action in the case of a typical financial requirement. Information gained by the bank in the process of relationship financing allows it to better predict the behavior and risk profile of the borrower. Combined with larger resources of information about the economic situation at the bank's disposal, it makes the bank more issues.

Benefits of information about customers, during the relationship financing, may be among the main factors of banks' profitability (Boot, 2000). Information gains during relationship banking allows banks to provide long-term, renewable and flexible financing to firms which are their customers, which decreases the probability of banks' bankruptcy due to conducting less risky activity (Keeley, 1990). The benefits for the bank depend on the quality of information about the customer and diversified quality of this information determines the level of the bank's specialization in establishing relational contracts. It was determined that having information about the behavior of customers provides significant benefits to both individual banks and entire banking systems (Cull, Haber and Imai, 2011).

Profit from relationship banking depends on the competition in the banking sector (Boot, and Thakor, 2000). Competitions increased in the banking sector decrease the profitability of relationship banking by lessen the intention to maintain close relationships (Ergungor and Thomson, 2005). However, this depends on the degree of market concentration (Elsas, 2005). Access to information about customers and its use may be the key to retain competitive advantage, especially for smaller banks (Berger and Udell, 2002; Boot and Marinc, 2008).

Benefits from relationship banking for firms are also broadly discussed in literature and concern mainly on reduction of the negative effects of information asymmetry (Diamond, 1991; Petersen and Rajan, 1994). Establishing a long-term relationship based on information collected by both parties reduces transaction costs resulting from negative selection and agency costs (Diamond, 1991; Boot, 2000). A strong relationship with a bank may also improve the firm's image and reputation.

A close bank-firm relationship, not only the bank monitoring costs would be at its lowest level, but does it suggest a solution to agency problem (Diamond, 1984). It can effective avoid wealth transfer from lenders to shareholders, and improve corporate governance (Limpaphayom and Polwitoon, 2004). Previous researchers also suggest that this relationship reduces the information asymmetries and incentive problems (Jensen and Meckling, 1976; Weinstein and Yafeh, 1998). It is largely agreed that such problems (asymmetry and incentive) affect corporate investment to a large degree. Therefore, firms with such a relationship should improve their access to capital and invest more in the market.

The strong bank-firm relationship maintains a significant role in the modern financial markets, especially during the financial crisis. Many researchers agree that burdens of debt

make companies more unstable in economic issue. A close bank relationship can reduce the costs of financial instability (Hoshi, Kashyap and Scharfstein, 1991). Moreover, it is possible that a firm's main bank is also the main bank of its suppliers and even its customers. Therefore, this relationship can overcome the issues of credit expansion and trustworthiness among customers, suppliers, firms and banks.

One of the advantages of lending relationships is it allow the lender to collect information about a borrower which is not easily reproduced by other financial institutions. This will give the lender a competitive advantage. Banks can know about a firm by observing it over time as well as over products. The bank can observe the firm's repayment history as well as other information about the firm and the manager which the bank obtains through the loan officer. History with the bank raises the banks expectation that the borrower is a good credit risk (Diamond, 1991).

In addition, relationships can be built through interaction over multiple products due to banks offers more than just lend money. They may manage the firm's cash account, factor its sales, and service its lock box. These additional services provide by the bank can give the bank an additional information on the current financial strength of the borrower and give an early indicator if the borrower experiences financial issue.

From firm's point of view, creating a good relationships with banks will help firms to improve business reputation, to reduce the leakage of information to competitors (Campbell, 1979), to decrease the negative impact of asymmetric information (Diamond, 1984 and 1991;

Fama 1985; Rajan, 1992; Holmstrom and Tirole, 1997; Bolton and Freixas, 2000), to reduce agency conflicts related to financial intermediation (Deloof and Vermoesen, 2010), to increase accessibility to loans, and to reduce the interest cost (Houston and James, 1996; Pertersen and Rajan, 1995). This will cause the firm not only depend on the cash flow liquidity within the firms so that the firms can easily invest in fixed assets with lower cost of capital, and reserves cash will be used to increase profitability (Fazzari et al., 1988; Hoshi et al., 1991).

Firm's investment is less sensitive to cash flow when a firm has a strong banking relationship (Shen and Wang, 2004). Diamond (1984) finds that a close relationship allows the bank to handle an active monitoring role which can lessen problems related to free-riding and information asymmetry. In addition, when firms establish close banking relationship through repeated lending from a bank, they will increase their reputation on the various funding sources.

Diamond (1991) also indicates that firms choose bank funding first in order to establish sufficient credibility and then access the capital markets. Research on Japanese firms shows that establishing good relationships with commercial banks increase accessibility to equity capital markets (Kutsuna, Smith and Smith, 2003). Furthermore, building up close relationships with banks helps firms overcome financial or business crisis (Hoshi et al., 1991).

These effects of relationships between banks and firms are likely to be more important for small firms due to small firm have a higher cost when obtaining investment funds from financial markets and rely heavily on banks as the main credit channel. Small firms tend to borrow from banks and to borrow from a few banks with which they have a long-term relationship. These relationships are an important feature of small business lending.

Small firms usually do not enter into contracts that are publicly visible because they do not have audited financial statements and consequently can have difficulty building reputations to signal high quality (Berger and Udell, 1998). Relationship lending enables banks to collect private information on the credit-worthiness of these firms since there may be little public information available on small firms (Strahan and Weston, 1998). These factors suggest that relationship lending may be particularly beneficial to small firms, including lower cost or greater availability of credit, protection against credit issue, and the provision of implicit interest rate or credit risk insurance.

A problem with the credit line from the bank can be considered as a bad signal about the firm for a small firm with a single relationship, even if the withdrawal of the credit is not linked to financial issue of the small business. As a result, small firms can have multiple banking relationships, which have higher transactions costs but also greater benefits than a single relationship (Berger and Udell, 1998).

#### 2.9.1 Exclusive Bank Relationship

A single bank with better information about a firm can impose hold-up costs that can adversely affect the borrowers' value (von Thadden, 1998). These additional costs can be reduce or eliminated by multiple banking relationships. The duration of a relationship between a firm and a bank also plays a role. During a recession, firms that have ongoing relationships with a bank are better able to obtain additional financing which allowing them to survive the recession with minimal loss (Longhofer and Santos, 2000).

There is another reason why firms prefer an exclusive financing contract, which is to reduce the costs of information disclosure. Campbell (1979) shows that the small companies with creative and interesting ideas are viewed as main participators. They do not have a lot of internal funds or external opportunities raising money. More importantly, they do not need to disclose the confidential information related to their product innovation to their competitors either in direct or indirect ways.

Banks provide firms with contracts that require high initial collateral together with interest payments that fall when the bank has verified the successful completion of financed projects (Boot and Thakor, 1994). This is particularly important for innovating and RandD-investing firms, as highlighted by Yosha (1995) and Bhattacharya and Chiesa (1995).

Previous research has largely discussed the importance of an exclusive banking relation. It is more advantageous for a company to have a single bank relationship rather than with several banking relations (Harhoff and Körting, 1998). The quality of information between the company and the bank is better shared due to there are comparative advantages of information for the two economic agents. It allows a better banking supervision.

Bank-firm relationship uniqueness allows a good impression about the debtor by the creditor as well as a confidence between the partners. The company finds that such a relation allows a firm to gain an easier access to the loans. Firms applying for loans will make it easier for the bank to evaluate the quality of their firms by providing an informational advantage (Boot, 2000). In return, the bank offers them the capital they need and assumes the risk relating to the process of financing.

Banking relations focus on the confidentiality of the information gathered by the banks. The main objective is to ensure an informational advantage where one should not reveal specific information about a company to its rivals. However, a context of information asymmetry, the bank is facing the risk of selection and consequently, it is difficult to differentiate the quality borrowers from non-quality ones. This can lead to a conflict situation between the two parties due to the insufficiency of the informational exchange. In order to avoid such conflicts, a good banking supervision is important to know potential borrowers.

The financial intermediation is a monitoring system where the exclusiveness in the banking financing constitutes a means of decreasing competition between banks (Diamond, 1984; Sharpe, 1990; Rajan, 1992). In fact, the banking monitoring is a complicated activity which includes the flows of treasury follow-up in the company and the renewals of the appropriation

(Nakamura, 1993). This explains the transmission of the investors' responsibility for the control of cash flows from the company to the bank. On another side, the financial intermediation represents in a certain way a mechanism of commitment in a relation where the exclusiveness in the financing is a tool which can reduce the costs of monitoring (Hellwig, 1991). Thus, the effectiveness of the banking intermediaries is increased within the frame work of the exclusive relation.

The increase in the performance of the firms is cause by the contact with the bank which makes it possible for a high probability of the success of the financed project. In fact, the evaluation of the customers' quality is based on an individual analysis which is determined by the exclusiveness in the banking relation and the co-operation between the bank and the company (Morck and Nakamura, 1999).

#### 2.9.2 Multiple Bank Relationship

Research on multiple relationships leads economists to consider other factors. A bank can acquire a great deal of unique information about a firm and the bank may be able to exploit this, when a bank and a firm have a long-term relationship. Long-lasting bank and firm relationships arise because high quality firms are unable to convey information about their quality to other banks (Sharpe, 1990).

Some banks choose have a relation with a company which already had other banking contacts in order to gain information on which loan- related decisions will be based (Japelli and Pagano, 2002). The improvement of the efficiency of the granted appropriations involves an improvement in the performance of company (Foglia, et al., 1998). The advantages of information sharing among banks are explained by Padilla and Pagano (1997).

Generally, the strategy of financing through a multiplicity of banking relations avoids the allocation of funds to non-profitable projects. Some firms choose to have multiple banking relations in order to minimize the probability of financing issue (Berger et al., 2005). When the banking system is fragile, the multiple banking relations took an insurance coverage against the loss of their in formational flows revealed by the company (Ongena et al., 2003).

The good quality company may find it beneficial to have several banks relations in order to limit the risk of misuse of the credit (Thakor, 1996). However, the dependency to the multiplicity of the banking relations presents a more advantageous solution, if the control was carried out by only one bank is ineffective. Moreover, the diversification of the banking relations does not necessarily shows an additional financing cost for the firm because it is dependent on the cost of control supported by the bank and of the facility of obtaining and data processing concerning the firms (Farinha and Santos, 2002).

From this review of the theoretical literature, it seems that the dependency to a multiple banking relation banks aims at avoiding the financing issue.

#### 2.10 Disadvantages of Lending Relationship

On the other hand, close cooperation with the bank may be costly for the firms (Sharpe, 1990; Rajan, 1992). Having information about the firms reduces competition in the market of bank loans, which causes an increase in loan prices in the future (Boot, 2000). The firms may reduce monopolistic pressure by comparing offers and actions of individual banks and establishing relationships with selected banks (Broecker, 1990). The costs of relationship banking are also borne by banks themselves.

In the case of relationship financing, banks should not continue to finance the firms which are experiencing financial difficulties although being familiar with the firm's situation. However, if the bank chose not to continue crediting, it would certainly lose the original loan (Boot, 2000). Relationship financing can also give rise to opportunistic behavior in firms, who take advantage of their creditworthiness and multi-annual cooperation with the bank in order to obtain further financing (Heffernan, 2007).

The relationship-building process between firms and banks is very costly to the borrowing side (Sharpe, 1990; Rajan, 1992). Since the lending side initiate the borrowing contract and gives the borrower financial aid at a favorable rate at the beginning, in return, the financial institution can accumulate a large amount of non-public material about the firm through this relation and finally have a bargaining power over the firm.

Furthermore, by using this monopoly power, a bank could even threat to cut off a firm's loan or simply charge it at a higher rate during the relationship process (Diamond, 1991). This is because there is an extra searching cost for the firm looking for an alternative bank to replace the current one. Others would view as a negative sign if a firm stops the relationship with its current bank, due to information gain and adverse selection problems (Castelli, Dwyer Jr. and Hasan, 2006).

Firms enjoy faster growth or high profitability among their peers if a close bank relationship helps firms' access to capital easier than those without such a relationship. However, some empirical evidence shows opposite stories. There are negative relationships between profitability and the degree of bank-firm relationship (Weinstein and Yafeh, 1998). They used sample period of the Japanese financial market liberalization from 1977 to 1986. The evidence shows that even though the strong bank relationship did increase access to funds, this phenomenon has stop since deregulation in early 1980s.

Meanwhile, they conclude that banks conduct rent-seeking behavior which means banks use their monopoly power to siphon profits from their client-firms. Further, they find that banks tend to shy away from risky but profitable investment and put pressure on their client-firms. This could be one of the reasons that companies with close bank ties cannot beat nonrelationship firms in the Japanese atmosphere. Moreover, Agarwal and Elston (2001) who examine a hundred large listed German firms from 1970 to 1986 do not find any evidence showing the benefits of German universal banking relationship. Nevertheless, there is a negative relation between interest payment and bank-influenced firms although it is only significant at the 10% level. In addition, Agarwal and Elston (2001) conclude that existing easy access to capital benefit could be balanced by less risky projects taking or by rent-seeking effect.

#### 2.10.1 Exclusive Bank Relationship

The exclusive banking relationship leads to the creation of a climate of trust between the two parties (Aya and Mouldi, 2012). However, this confidence can be fully used by the bank and thus spread the information collected with a third party. The exclusive relation sometimes can be harmful. There is a risk that private information will be reveal to a third party since the exclusive relation requires the exchange of information between the bank and its customer (the suppliers, the customers or the competitors) during the activities of the bank council.

The probability of transferring information about a company can be increased if the bank maintains a relation with competitive companies (von Rheinbaben and Ruckes, 2004). Therefore, in the case of banking concentration, the firms' performances reduce when the probability that the companies are in relation to the same bank is increased (von Rheinbaben and Ruckes, 2004). The information confidentiality is important in particular when it comes to strategic nature.

Furthermore, the exclusive banking relation can harm the quality of bank control. The banking supervision is started only if the bank estimates to optimize its profits following the financing of the company. The bank will be strongly involve in the choice of the projects to finance, the selection of investment and the choice of financing channel when they have private information about the firm (Bellouma, 2005).

A poor quality firm (bad reputation and firm having financial problems) may not find it beneficial to involve itself with a multiple banking supervision. The uniqueness of the relation is the bank can be pushed to reduce its monitoring and to reduce its techniques of monitoring (Udell, 1989). The firms tend to take more chances because they know that they will be financed in the case of financial problems. The exclusive banking relation leads to a fraudulent diversion where the funds used other than supporting the reproduction of the personal information, which resulting a cost of misuse of the credit and reducing the firms' performance (Berger and Udell, 2002).

#### 2.10.2 Multiple Bank Relationship

Having several banking relations does have negative effects on the firms. The multiplicity of banking relations reduces the possibility of bank to collect information and to control the company. The multiple banking relations show the appearance of a similar behavior in the banks (Petersen and Rajan, 1995). Indeed, each bank tries to reduce its effort of evaluation and to base its own decision on that compare to other bank (Rajan, 1992). Banks can grant credits by requiring high guarantees of carrying out their activities of control when faced with multiple relations.

The firms tend to takes the chance to misuse the banks financing without assistance, advice and follow-up, by widening its banking relations. The firms tend to refund a loan contracted at a bank by another granted by other bank without control. In other words, the firms can require funds higher than its real need and use the additional funds in risky activities. This behavior decreases the feeling of its real risk and damages its performance (Foglia et al., 1998).

Moreover, the maintenance of several banking relations by the firms is proof by the refusal of the initial bank to present the required funds to it (Harhoff and Körting, 1998). This idea was supported by Farinha and Santos (2002) who show that firms at significant risk of credit borrow from several banks. Therefore, this kind of banking strategy increases the probability of turning down loan applications presented by firms (Child and Terence, 2005).

# Chapter 3 Methodology

# 3.1 Introduction

This chapter discusses the theoretical framework followed by hypothesis development, data collection and variable measurement, as well as model specification and analysis in the last section. This chapter also explains the research design conducted and how the data was collected. Data are very important in guaranteeing and ensuring the reliability and accuracy of the findings in a study. This chapter also clarifies the method used in analyzing the relationships between bank-firm relationship and firm performance of top 100 listed companies in Malaysia.

#### **3.2** Theoretical Framework

The firm performance in this study is measured by two different approaches; finance or market based approach (Tobin's Q) and accounting based approach (ROA and ROE). The independent variables used in this study is number of bank relationship and credit financing, while the control variables is firm size and foreign ownership. The independent variables chosen due to the ability to examine the impact of bank –firm relationship to the firm and also to measure effectiveness of a firm using its credit financing. The control variable is chose due to the small impact on the effectiveness of the firm operating business.

Research model:

#### Figure 3.1 Research Framework



# 3.3 Hypothesis Development

The research hypotheses are proposed below according to research questions in chapter one and relevant literature reviewed in chapter two.

- H1 There is a significant relationship between the number of bankfirm relationships and firm performance.
- H2 There is a significant relationship between short-term credit financing of a firm and its performance.
- H3 There is a significant relationship between long-term credit financing of a firm and its performance.

- H4 There is a significant relationship between firm size and firm performance.
- H5 There is a significant relationship between foreign ownership in a firm and its performance.

#### **3.4 Data Collection**

#### 3.4.1 Sample Selection

The initial sample of the study consists of the Top 100 public listed companies in Malaysia, in the year of 2013. These Top 100 companies were selected to be examined because their market capitalization formed about 85% of the total market capitalization of companies listed in Bursa Malaysia. (Azlan, Jamal, Geetha, Mohidin, A.Karim, Lim, Ch'ng (2013)). The results from the study are thus expected to be able to generalize the listed companies on Bursa Malaysia as far as firm-bank relationship is concerned.

In this study, financial institutions such as investment funds, insurance firms, REITS, Trusts firms, banking and finance and securities firms are excluded from the sample due to the differences in characteristics in their assets and capital structure, compared to non-financial firms. After excluding the 19 firms that fall under the criteria mentioned earlier, and excluding 3 firms which merged and were de-listed, there left a final total of 78 firms, representing a sample rate of 78% of the population for analysis. All the sample data are categorized into groups the same as in Bursa Malaysia as follow: trading and services, industrial product, plantation, construction, consumer, properties and hotel.

The data collected in this research comprise two categories: dependent variables and independent variables. There are three dependent variables in this research based on two different measurements: value-based measure (Tobin's Q), and accounting measures (return on assets, ROA, and return on equity, ROE). Tobin's Q is used to test how the bank firm relationship affects the firm market performance. Firms' profit is measured by ROA (return before tax on assets) and ROE (return after tax on equities).

The independent variables consist of the number of bank relationships, short-term credit financing relationships, long-term financing relationships and overall credit financing relationships. Two firms' characteristics variables (firm size and foreign ownership) are included in the examination as control variables. All the data needed in this research are extracted from respective corporate annual reports of the firms.

#### 3.4.2 Ordinary Least Square Regression and Model Specification

In this study, the method of ordinary least squares (OLS) regression is used to examine the relationships between firm performance and number of banking relationships, credit financing relationships (short-term, long-term and overall), and two control variables (firm size, and foreign ownership).

In analyzing firm performance and firm-bank relationships with firm performance, OLS is the common method used by the previous studies. The model developed and used in this study represents the appropriate assumptions derived from previous literature to show the relationships between the dependent variables and the independent variables.

# 3.4.2.1 Model Specification

The general structural equation used in this study to test and explain the relationship is as follows:

ROA = 
$$\beta_0 + \beta_1 BNR + \beta_2 STF + \beta_3 LTF + \beta_4 FZS + \beta_6 FOW + \varepsilon$$
 (1)

ROE 
$$= \beta_0 + \beta_1 BNR + \beta_2 STF + \beta_3 LTF + \beta_4 FZS + \beta_6 FOW + \epsilon$$
(2)

Tobin's Q =  $\beta_0 + \beta_1 BNR + \beta_2 STF + \beta_3 LTF + \beta_4 FZS + \beta_6 FOW + \epsilon$  (3)

Where:

ROA/ROE/Tobin's Q	-	Firm's performance
β	-	Intercept of the equation
BNR	-	Number of firm-banking relationships
STR	-	Short-term credit financing relationships
LTR	-	Long-term credit financing relationships
FSZ	-	Firm size
FOW	-	Foreign ownership
3	-	Error Term

# **3.5** Research Model and Measurement of Variables

#### **3.5.1 Dependent Variables**

Tobin's Q is a ratio of the combined market value of all the companies on the stock market which should be about equal to their replacement costs. The ratio is devised by James Tobin from Yale University (Tobin, 1969; Brainard and Tobin, 1968). The Q ratio is calculated as the market value of a company divided by the replacement value of the firm's assets:

Q Ratio = Total Market Value of Firm/Total Asset Value

Where:

- $\circ$  Total Market Value of Firm = (Share price) x (outstanding shares)
- Total Asset Value = Current Asset + Non-current Asset

Tobin's Q Ratio provides information on how well a company's investments pay off. A low Q (between 0 and 1) means that the cost to replace a firm's assets is greater than the value of its stock. This implies that the stock is undervalued.

The other dependent variables for measuring firm performance are ROA and ROE.

ROA = Net Income / Total Assets ROE = Net Income / Shareholder's Equity

#### 3.5.2 Independent variables

For defining the independent variables and control variables, Table 3.2 summarizes the measurement description and the prediction of the result. Most measurements and expected relations are consistent prior with previous research (indicated in italics in the table). There are three independent variables and two control variables used in this research. The independent variables are as follow:

#### (i) Number of bank relationships (BankN):

This variable has quadratic function relation with dependent variable (Castelli et al., 2006). This means that when a firms is involved in more number of bank relationships, the firm's performance will increase due to the increase in the power of negotiation, hold-up cost reducing, ease of getting more loans for investment, and liquidity of cash flow improved.

However, if a firm is involved in many number of bank relationships, they will also have to bear extra transaction costs and representative costs. The firm's profit will thus decrease because of those costs. Firm's performance will therefore be negatively affected.

#### (ii) Short-term credit financing (STF):

This variable is measured by short-term loan over total loan. This variable is expected to have negative impact on firm's performance because of several factors, for example time constraint and higher interest rate and hold-up cost.

(iii) Long-term credit financing (LTF):

This variable is measured by short-term loan over total loan. This variable is expected to have positive impact on firm's performance due to the ability to fully used the financing despite from the longer period given. It also means that the firm has stronger long-term credit financing relationships than others.

Apart from the four independent variables mentioned above, two control variables are included, as follow:

(i) Firm size:

This variable is measured by the logarithm of total assets (LAsset). This variable is expected to positively relate to firm's performance because of following reasons:

- Small firms are less likely to be close to the minimum efficient scale needed to operate efficiently in a market (Audretsch and Mahmood, 1994; Geroski and Mata, 2005).
- Small firms apply different technologies based on their size. Due to their small scale, they use less capital intensive approaches, so variable costs achieve a higher ratio within the total costs of the firm.

(ii) Foreign ownership:

Wei, Xie and Zhang (2005) and Tian and Estrin (2008) suggest a positive relationship between firm performance and foreign shareholdings. Foreign ownership is measured by the percentage of share own by the foreign. They may have access to cheaper financing via the parent firm or by virtue of their implicit government guarantee. They note that foreign investors can effectively monitor and influence management of the firm that they are holding. Thus, a positive relationship between firm value and firm's foreign ownership is expected.

Table 3.2 demonstrates the description of the independent and control variables used in the study. The prediction of the directions (impact on firm performance) is stated based on past literature.

Test Variables	Description of Measurement	Prediction of Direction
BankN	No. of banking relations in the firm	-
	Aya and Mouldi (2012)	
STF	No. of short-term credit financing in the firm	-
LTF	No. of long-term credit financing in the firm	
	Thanh and Minh Ha (2013)	+
Control Variables	Description of Measurement	Prediction of Direction
Firm Size	Log of Asset and Log of Turnover	+
	Cao, Chen and Chi (2010)	
Foreign	Percentage of foreign ownership in the firm	+
ownership	Ghosh (2004)	

Table 3.1: Ir	ndependent and	<b>Control Variables</b>
---------------	----------------	--------------------------

# Table 3.2: Description of Variables Used in the Study

Dependent Variable	Description
Tobin's Q	Market value of a company divided by the replacement value of firm's assets
ROA	Ratio of net income scaled by total asset
ROE	Ratio of net income scaled by shareholders' equity
Independent Variable	Description
No of bank relationships (BankN)	Total number of banks that lend loans to a firm in fiscal year 2013
Short-term credit financing (STF)	Ratio of short-term bank loans to total loan of a firm
Long-term credit financing (LTF)	Ratio of long-term bank loans to total loan of a firm
Control Variable	Description
Firm size (LAsset)	Natural logarithm of total assets
Foreign ownership	Percentage of foreign ownership of the firm

# Chapter 4 Results and Findings

### 4.1 Introduction

In this chapter, the results are analyzed and discussed. Data collected from company's annual reports were run using several analyses, which are descriptive analysis, multicollinearity test, correlation analysis, and linear regression analysis, by using Statistical Package for the Social Science (SPSS) version 16.0. The purpose of this chapter is to present and discuss the findings of this study. The chapter is organized into sections as follow: Section 4.1 presents the introduction, Section 4.2 presents the descriptive analysis, Section 4.3 examines the descriptive statistics, Section 4.4 shows the correlation among independent variables using a Pearson product moment correlation analysis, and Section 4.5 presents the results of regression analysis of the model tested.

# 4.2 Descriptive Analysis

Table 4.1 shows the distribution of companies by sector that is used for analysis in this study. The final total number of companies used as empirical evidence in Malaysia is seventy eight. These sample companies have been organized for industry differences that may impact the accruals behavior by firstly estimating the discretionary accruals according to industry–year portfolio, and secondly by involving industry dummies in the regression. From the 81 companies, three were further excluded because of them being delisted or merged during the observation year.

#### **4.2.1. Sector Classification**

These 78 companies are divided into 7 sectors according to the classification by Bursa Malaysia. Table indicates that trading and services sectors have large sample (36 companies), industrial product (9 companies), plantation (10 companies), construction (5 companies), consumer (11 companies), properties (6 companies) and hotel (1 company).

Sector	Number of companies	Percentage (%)
Trading and services	36	46.2%
Industrial product	9	11.5%
Plantation	10	12.8%
Construction	5	6.4%
Consumer	11	14.1%
Properties	6	7.7%
Hotel	1	1.3%
Total of companies	78	100%

 Table 4.1: Sample Distribution of Companies Based on Sector Classification

#### 4.2.2 Number of Bank Relationships

Table 4.2 shows the percentage of bank number relationship of the sampled companies from all sectors. The range of percentage is divided into three proportions as follow:

1. First proportion (0 - 10 relationships)

- 2. Second proportion (11-20 relationships)
- 3. Last proportion (more than 20 relationships)

Table 4.2 shows that majority of the companies (71 companies or 91%) has a proportion of less than 10 bank-firm relationships, 6% have between 11-20 bank-firm relationships, while there is only one company that has more than 20 bank-firm relationships.

No of Bank Relation	No of Firm	Percentage
0-10	71	91.03
11 - 20	6	7.69
> 20	1	1.28
Total	78	100.00

Table 4.2: Percentage of Bank Number Relationships of Sampled Companies

#### 4.2.3 Percentage of Foreign Ownership of Sample Companies

Table 4.3 shows the percentage of short-term credit financing relationship of the sample companies from all sectors.

Foreign Ownership	No of Firm	Percentage
0 - 30	71	91.03
31- 50	1	1.28
> 50	6	7.69
Total	78	100.00

 Table 4.3: Percentage of Foreign Ownership of Sampled Companies

#### 4.3 **Descriptive Statistics**

Table 4.4 presents a summary of the dependent variables and independent variables using descriptive statistics that comprises data for mean and standard deviation. It is divided into four sections: minimum value, maximum value, mean and standard deviation. It includes independent and control variables as tested variables. The results show that the mean for ROA ratio for all 78 companies is 6.5 percent and the minimum and maximum of ROA is 0 and 13.7 percent respectively. The minimum and maximum of the ROA indicates how profitable a company is relative to its total assets. The mean for ROE ratio is 12.3 percent and minimum and maximum of ROE is 0 and 27.1 percent respectively. This ratio indicates how profitable a company is by comparing its net income to its average shareholders' equity.

Meanwhile, the mean for Tobin's Q ratio is 0%, minimum and maximum of Tobin's Q is 0 and 0.4 percent respectively. A low Q (between 0 and 1) means that the cost to replace a firm's assets is greater than the value of its stock. The impact of bank-firm relationships to firm performance is estimated using independent variable which is short-term credit financing (STF) and long-term credit financing (LTF). This shows how much the firm depends on the type of credit financing, either short-term or long-term. The mean for STL is 41.9 percent, while the mean for LTF is 55.3 percent. The minimum and maximum for both variables is 0 and 1. The mean for firm size (LAsset) is 6, which indicates that the average Malaysian firms' size is moderate because the value for minimum size is 5 and the maximum value is 8. The mean for FOW (foreign ownership) is 15%, minimum is 0.65% and maximum is 74%, indicating the percentage of foreign ownership in the sample data.

	Minimum	Maximum	Mean	Std. Deviation
ROA	.0027	.1371	.0651	.0315
ROE	.0064	.2714	.1203	.0509
TQ	.0001	.0041	.0016	.00096
BNR	.0000	23.0000	5.9615	4.1417
STL	.00	1.00	.4194	.3706
LTL	.00	1.00	.5536	.3745
LAsset	5.58	8.65	6.7125	.5755
FOW	.65	74.33	15.0064	16.5352

**Table 4.4: Descriptive Statistic of Continuous Variables** 

#### 4.4 Multicollinearity

Multicollinearity is a statistical phenomenon when two or more variables in a regression model are highly correlated. The multicollinearity test is an important analysis because the existence of muticollinearity shows a critical issue on the regression model due to the obstacles that occurred when identifying the consequence between independent variables and dependent variable.

Multicollinearity is one of the many ways that can be used to check the abnormal relationships that might exist among the independent variables tested because most of the variables usually explain the result which variables are affected to be established in the study. In order to detect multicollinearity and to measure the results, Variance Inflation Factor (VIF) is used. In the situation where the Variance Inflation Factor (VIF) is above 10, the independent variables in the study are considered as highly correlated. Thus, when running the multiple regression models, the command for multicollinearity diagnostics to include VIF is selected in the analysis. In Table 4.5, results show that there is no multicollinearity problem with the independent variables because all the variables have VIF values of below 10.

		Collinearity Statistics		
Model		Tolerance	VIF	
1	BNR	.855	1.170	
	STL	.169	5.910	
	LTL	.163	6.134	
	LAsset	.907	1.102	
	FOW	.900	1.111	

 Table 4.5: Multicollinearity Test Summary

# **4.5 Correlation Analysis**

Correlation analysis is used as a statistical tool analysis in order to determine the relationship level of one variable to another. Before carrying out the linear regression and in order to establish the association between the dependent and independent variables, a correlation matrix is developed. Table 4.6 shows the correlation relationships between firms' performance (ROA, ROE and Tobin's Q), independent variables and control variables. The highest correlation among the independent variables is between LTF and BNR, which is 0.339 (p<0.05), suggesting that the higher the number of bank relationships, the more will be the number of long-term credit financing. The result shows a negative relation between LTL and STL (-0.909) (p<0.05). The number of bank relations is negatively correlated to firm's performance as indicated by the negative values of 0.356, 0.299 and 0.332 respectively. Foreign ownership is found to have has a positive effect on firm performance.

	ROA	ROE	TQ	BNR	STL	LTL	LAsset	FOW
ROA	1							
ROE	.741**	1						
TQ	.442**	.482**	1					
BNR	356**	299**	332**	1				
STF	$.280^{*}$	.152	.185	311**	1			
LTF	227*	111	169	.339**	909**	1		
LAsset	068	.037	013	.176	176	.253*	1	
FOW	.071	.011	.023	231*	.271*	253*	114	1

**Table 4.6: Correlation Matrix Summary** 

\*\*\*, \*\* and \* indicates significance at 1%, 5% and 10% levels

# 4.6 **Regression results**

The effects of the independent variables on firms' performance (measured by ROA, ROE, and Tobin's Q) are described in Table 4.7 as follow:

	ROA	ROE	Tobin's Q	
R-squared	.173	.116	.127	
Adjusted R-Squared	.115	.054	.066	
Std. Error of the Estimate	.0296	.0495	.00093	
Ν	78	78	78	

**Table 4.7: Linear Regression Model Summary** 

Table 4.7 shows the results from the regression model. The R squared ( $R^2$ ) and adjusted R square are used as the explanatory model. The  $R^2$  in the table explains by percentage the influence of the independent variables on the dependent variables. The table shows that 17.3 percent (ROA), 11.6 percent (ROE) and 12.7 percent (Tobin's Q) respectively of the variation is the dependent variable (s) is explained by the independent variables. The other remaining percentage of dependent variable(s) variation could be explained by other factors.

The lower result of the  $R^2$  in ROA, ROE and Tobin's Q is considered acceptable for this kind of research because previous studies have obtained lower results. (See for example, Castelli, Dwyer, and Hassan, 2006). The effects of the independent variables on firms' performance (measured by ROA, ROE, and Tobin's Q) are described in Table 4.8 (a), (b), and (c) as follow:

Table 4.8 (a) shows the results of the coefficient linear regression for the present study. The number of banking relationship affects firm performance negatively and significantly at a significant level of 5%. Short-term financing effect firm performance significantly and positively at significant level 10%. They indicate that if the number of bank relationship increase by one unit, the ROA for the companies is decreased by 0.002. For short-term financing, if it is increased by one, the ROA is increased by 0.037. For long-term financing, if it is increased by one, the ROA is increased by 0.022. There are no changes in ROA if the firm size and foreign ownership increase by one.

	Unstandardized	Coefficients	Standardized Coefficients		
Model	В	Std. Error	Beta	t	P-value
(Constant)	.056	.043			
BNR	002	.001	322	-2.776	.007**
STF	.037	.022	.436	1.675	.098*
LTF	.022	.022	.267	1.006	.318
LAsset	.000	.006	008	072	.943
FOW	.000	.000	055	484	.630

Table 4.8 (a) Linear Regression Analysis

(Dependent Variable: ROA)

\*\*\*, \*\* and \* indicates significance at 1%, 5% and 10% levels
When firm performance us measured by ROE, the result show that the number of banking relationships is negatively and significantly correlated with firm performance at significant level 1%. All other variables (short-term financing, long-term financing, assets and foreign ownership) are positively correlated by none of them is significantly related. Of these four variables, STF affects ROE the most. For short-term financing, if it is increased by one, the ROE is increased by 0.039. For long-term financing, if it is increased by one, the ROE is increased by 0.029. The ROE increases by 0.07 if the firm size increased by one unit. Foreign ownership is found to have no effect on ROE.

Table 4.8 (b) Linear Regression Analysis

	Unstandardized	Coefficients	Standardized Coefficients		
Model	В	Std. Error	Beta	t	P-value
(Constant)	.068	.072		.942	.349
BNR	004	.001	315	-2.631	.010***
STF	.039	.037	.281	1.043	.301
LTF	.029	.037	.213	.775	.441
LAsset	.007	.010	.079	.682	.498
FOW	.000	.000	075	641	.524

(Dependent Variable: ROE)

\*\*\*, \*\* and \* indicates significance at 1%, 5% and 10% levels

Table 4.8 (c) shows the results of the coefficient linear regression for the study. The number of banking relationship affects firm performance negatively and significantly at a significant level of 1%. Short-term financing and long-term financing have no effect on Tobin's Q, while assets have positive but very low impact on Tobin's Q. Foreign ownership effects negatively Tobin's but the impact is very low.

(Dependent Variable: Tobin's Q)								
	Unstandardized	Coefficients	Standardized Coefficients					
Model	В	Std. Error	Beta	t	P-value			
(Constant)	.001	.001		1.018	.312			
BNR	00007634	.000	328	-2.750	.008***			
STF	.000	.001	.175	.655	.514			
LTF	.000	.001	.070	.255	.799			
LAsset	.00008253	.000	.049	.426	.672			

.000

-.077

-.661

.511

Table 4.8 (c) Linear Regression Analysis

\*\*\*, \*\* and \* indicates significance at 1%, 5% and 10% levels

-.000004476

FOW

## 4.7 Discussion

#### Impact of the number of bank relationships (BankN) on firm performance

Castelli et al. (2006) find a negative influence between number of bank relationships and firm performance, while Garriga (2006) points out the opposite. The estimation results of this present study for all the three performance measures (ROA, ROE and Tobin's Q) are similar to Castelli's.

The increase in the number of bank relationships helps firms to increase in number of credit supply resources and solve hold-up problems. However, the firms may have to bear increased transaction costs, representative cost and free-riding problems. An increase in the number of banks in banking relationship helps firms to easily find more sources of funding, but this also makes firms easily use credit provided to invest in unnecessary business activities. When a firm get credit financing from a bank, the process of applying credit financing from other banks will be easier since the other banks will rely on the evaluation from the previous banks.

#### Impact of short-term credit financing on firm's performance

Hiraki et al. (2003) find that short-term credit financing has a negative relationship to firms' performance by using a measure of ratio of total main bank loans to total liabilities in estimating their models. Firms which have strong short-term credit financing relationships are expected to be less effective than the others which do not have such relationships. Short-term credit financing helps firms solve financial problem in short-term, but firms have to accept

higher borrowing cost in exchange. There are also some firms that use short-term loans to cover to long-term businesses. The explanation causing the firm to use short term relation as mentioned in the review of literature shows the reason for the negative impact of short-term financing relationship to firm performance.

This present study shows that short-term credit financing have a positive but not significant relationship with ROA and ROE, and no effect on Tobin's Q. Theoretically, when firms establish strong short-term credit financing, they would have an opportunity to achieve more effectively short-term business activities; thereby this affects positively overall firm performance (Baum, Sch<sup>°</sup>afer and Talavera, 2007).

#### Impact of Long-Term Credit Financing on Firm Performance

This variable shows that even though there is no significant relation with any of the firm performance measures (ROA, ROE and Tobin's Q, there is a positive relationship between long-term credit financing and ROA and ROE, while there is no impact on Tobin's Q. The current research finds that firms having strong long-term credit financing relationships are more effective compared to other firms (Asterbro and Bernhardt, 2003). This suggests that concentrating on long-term growth through long-term investment can help firms create sustainable performance.

However, bearing long-term financing might make the firm face higher total interest compared to total interest in the short-term credit financing. Some firms might also misuse the financing in non-profitable aspect. It can be concluded that even though firms having strong long-term credit financing relationships can use financial leverage (in long-term loans) better, but the effectiveness of using total assets is not significantly better.

#### Impact of firm size on firm performance (LAsset)

This study shows no significant result between firm size and firm performance. However, it positively correlated firm performance. Total assets impact negatively all the three measures of firm performance, implying that firms with more assets tend to have lower firm performance compared to others. This finding is consistent with those of Thanh and Min Ha (2013), and Garriga (2006).

#### Impact of foreign ownership (FOW) on firm performance

Foreign ownership has no significant relationship on firm performance. However, there is a positive relationship with firm performance as measured by ROA and ROE. This finding is consistent with a previous study by Azzama, Fouad, and Ghosh, D. K. (2013) that indicates that there is a positive relationship between foreign ownership and firm performance. When foreign investor owns shares in a company, they can help improve the company's access to the capital market.

# Chapter 5 Summary and Conclusion

## 5.1 Introduction

The purpose of this chapter is to represent the findings with discussion about the contribution and limitations of this study, as well as suggestion for future research. The final chapter is organized in several sections as follow: Section 5.1 is the Introduction, Section 5.2 summarizes general findings of this study, Section 5.3 shows the implication and limitations, Section 5.4 provides suggestions for future research, and Section 5.5 concludes the study.

# 5.2.1 Summary of General Findings

This study examines the effect of bank-firms relationships on firm's performance for a sample of public listed companies in Malaysia, by investigating the impact of three bank-firm relationship variables (number of bank relationships, short-term credit financing, long-term credit financing) and two firm characteristics (assets and foreign ownership) on three measures of performance (ROA, ROE, and Tobin's Q).

The general findings in this study indicate that two variables (BNR and STF) have significant relationship on firm performance. The number of bank relationship (BNR) shows a significant negative relationship on the three firm performance measures (ROA, ROE, and Tobin's Q). While short-term financing shows a significant positive relationship on ROA, and a positive effect on ROE, it has no impact on firm performance measured by Tobin's Q. Other variables namely long-term financing, firm size and foreign ownership shows no significant relationship with firm performance. Foreign ownership shows positive impact on ROA and ROE but is negatively correlated with Tobin's Q.

Based on the research findings presented in chapter four (results and findings), the hypotheses developed have been tested to determine whether they can be accepted or otherwise.

Hypothesis	Result
	<u> </u>
• H1 - There is a significant relationship	Supported
between the number of bank-firm relationships	
and firm performance.	
• H2 - There is a significant relationship between	Supported (for ROA)
short-term credit financing of a firm and its	
performance.	
• H3 - There is a significant relationship between	Not Supported
long-term credit financing of a firm and its	
performance.	

#### Table 5.2 Acceptability of Hypothesis

•	H4 - There is a significant relationship between	Not Supported
	firm size and firm performance.	
•	H5 - There is a significant relationship between foreign ownership in a firm and its performance.	Not Supported

# 5.3 Limitations of the Study

There are two limitations of the study as listed below:

- 1. Sample data used in this study covers a period of only one year that is 2013, as there are several companies yet to publish their annual reports for the financial year of 2014. The study period of 2013 can be considered as appropriate since it represents the latest year of which all annual reports for the companies sampled were available.
- 2. The samples of firm is derived from the Top 100 public listed companies in Malaysia which represent the larger firms among all the public listed companies in Malaysia. Top 100 companies are chosen because of the limited time to analyze more than 1000 companies listed in Bursa Malaysia.

## 5.4 **Recommendations for firms**

Based on the conclusion, there are some recommendations as follows:

- Firms should focus on quality of banking relationships rather than trying to establish more number of relationships. In distress period, establishing more number of bank relationships helps firms increase their credit supply resources and solving the holdup problem, but firms need to bear the increase of transaction costs, representative cost and free-riding problem.
- 2) Firms should increase their short-term credit financing if there are more opportunity in short-term business activities rather than long-term business activities. This will help increase their firm's profit. Once firms overcome the pressure of short-term performance through improving managerial activities, firms will have sufficient time concentrating on improving long-term business activities radically. This helps firm sustainably grow and earn higher profit
- 3) Banks should avoid opportunities to increase more number of bank relationships. This increase is likely that firms are looking for more sources of credit financing to solve short-term problem in the distress period rather than of solving business activities. Banks should also consider the offer of easy financing credit which is financed credit from previous banks.

4) Firms may carefully consider increasing foreign ownership rate in firms because in Malaysia, the firm with less foreign ownership tends to get credit financing easier compared to firms with more foreign ownership, except for those companies which are franchisee, for example Nestle, Carlsberg and British American Tobacco, which are owned by foreign firms

# 5.5 Recommendations for future research

Future research can be developed in several areas as list below:

- This study relies mainly on quantitative analysis as based research approach. Future research might follow up with face to face interview with the person in charge of the selected companies that been used as the sample since they did not disclose everything in the annual report, such as name of the borrowing banks.
- 2. The empirical study of this thesis focused only one year, 2013. In future research, a longer time frame should be carried out to get more accurate results and findings.
- 3. The sample used in this study is based on Top 100 public listed companies in Malaysia, which is among the largest companies in Malaysia. In future study, the samples should cover also small and medium companies in order to balance the sample data.

#### 5.6 Conclusion

From the results and findings in the earlier chapter, it can be concluded that there are significant impact of number of bank relationship and short-term financing with firm performance. This study shows that if firms can establish a strong short-term credit financing, there a positive impact on their firm performance.

However, there result shows no significant relationship between long-term financing, firm size and foreign ownership. If the firm sustains long-term credit financing, there are positive impact on their performance. It is not appropriate for firm to focus on short-term relationship due to time constraints when dealing with the short period given to fully utilize the said credit financing. Focusing on long-term relationships helps firm get better performance, since there are no time constraints that need to be monitor when utilizing the credit financing.

The increase in total assets during this period through borrowing from banks reduces the effective use of assets but improves financial leverage and these lead to increase in effectiveness of using equity. The use of tangible assets through debt financing from financial institutions is ineffective leading to reduction of firms' performance.

The foreign ownership shows no significant relationship but a positive impact on firm performance. This is indicates that the foreign ownership will have influence in the firm decision making and operating business. Therefore, it will help to increase the firm performance.

## References

- Agarwal, R., and Elston, J. A. (2001). Bank-firm relationship, financing and firm performance in Germany, *Economic Letters*, 72(2), 225-232.
- Angelini, P., Di Salvo, R., and Ferri, G. (1998). Availability and cost of credit for small businesses: Customer relationships and credit co-operatives. *Journal of Banking Finance*, Vol. 22, 925–954.
- Asterbro, T., and Bernhardt, I. (2003). Start-up financing, owner characteristics, and survival. *Journal of Economics and Business*, 55, 303–319.
- Audretsch, D. B., and Mahmood, T. (1994). The rate of hazard confronting new firms and plants in US manufacturing. *Review of Industrial Organization*, 9(1), 41-56.
- Aya, A., and Mouldi, D. (2012). Banking relationship's number, firm credit and corporate performance in Tunisia. *Economics and Finance Review*, Vol. 2(2), 26-36.
- Azlan, A., Jamal, A., Geetha, C., Mohidin, R., A.Karim, M. R., Lim, T. S., Ch'ng, Y. (2013). Capital Structure Decisions: Evidence from Large Capitalized Companies in Malaysia, *Journal of Contemporary Research in Business*, Vol. 5, 5.
- Azzama, I., Fouadb, J., and Ghosh, D. K. (2013). Foreign ownership and financial performance: Evidence from Egypt. *International Journal of Business*, 18(3), 233-254.
- Baum, C. F., Schäfer, D., and Talavera, O. (2006). The effects of short-term liabilities on profitability: the case of Germany (No. 635). DIW-Diskussionspapiere.
- Bellouma, M. (2005). Trade credit policy and agency theory: Evidence from Tunisian export companies.
- Berg, G., and Schrader, J. (2009). Access to credit, natural disasters, and relationship lending. Journal of Financial Intermediation, Vol. 21 (4), 549-568.
- Berger, A. and Udell, G.F. (2002). Small business credit availability and relationship lending: The importance of bank organisational structure. *Economic Journal*, *Vol.* 95, 1165-1183.
- Berger, A. N., and Udell, G. F. (1998). The economics of small business finance: The roles of private equity and debt market in the financial growth cycle. *Journal of Banking and Finance, Vol.* 22, 613-673.
- Berger, A., Miller, N.H., Petersen, M.A., Rajan, R.G., and Stein, J.C. (2005). Does function follow organizational form? Evidence from the lending practices of large and small banks. *Journal of Financial Economics*, Vol.76, 237-269.

- Bernanke, B. S., and Gertler, M. (1995). Inside the black box: The credit channel of monetary policy transmission. *Journal of Economic Perspectives*, *Vol.* 9, 27–48.
- Bhattacharya, S., and G. Chiesa, (1995). Financial intermediation with proprietary information. *Journal of Financial Intermediation* 4, 328-357.
- Bhattacharya, S., and Thakor, A. V. (1993). Contemporary banking theory. *Journal of Financial Intermediation*, Vol. 3, 2–50.
- Bolton, P., and Freixas, X. (2000). Equity, bonds, and bank debt: Capital structure and financial market equilibrium under asymmetric information. *Journal of Political Economy*, Vol. 108, 324-351.
- Bolton, P., and Scharfstein, D. S. (1996). Optimal debt structure and the number of creditors. *Journal of Political Economy*, 104, 1-25.
- Boot, A. W. A. and Marinc, M. (2008). Crisis management and lender of last resort in the European banking market. 233-250. P. Alessandrini et al. (eds.). The Changing Geography of Banking and Finance, DOI 10.1007/978-0-387-98078-2 11, C Springer Science + Business Media, LLC 2009.
- Boot, A. W. A. (2000). Relationship banking: What do we know? *Journal of Financial Intermediation*, Vol. 9, 7-25.
- Boot, A. W. A., and Thakor, A. V. (2000). Can relationship banking survive competition? *Journal of Finance, Vol. 55*, 679-713.
- Boot, A. W. A., and Thakor, A. V. (1994). Moral hazard and secured lending in an infinitely repeated credit market game. *International Economic Review*, 35, 899–920.
- Boyd, J. H., and Prescott, E. C. (1986). Financial intermediary coalition. *Journal of Economic Theory, Vol. 38*, 211–232.
- Brainard, W. C., and Tobin, J. (1968). Pitfalls in financial model building. *American Economic Review*, 58 (2), 99–122.
- Broecker, T. (1990). Credit-worthiness tests and interbank competition. *Econometrica*, Vol. 58, No. 2, (Mar), 429-452.
- Campbell, T. S. (1979). Optimal investment financing decisions and the value of confidentiality. *Journal of Financial and Quantitative Analysis*, 14, 232-257.
- Cao, W., Chen, J., and Chi, J. (2010). Bank firm relationship and firm performance under a state-owned bank system: Evidence from China, *Banks and Bank Systems*, *Vol. 5*, Issue 3.
- Carpenter, M. A and Westphal, J. D (2001). The strategic content of external network ties: examining the impact of director appointment on board involvement in strategic decision making. *Academy of Management Journal*, 44, 639-660.

- Castelli, A., Dwyer, G. P. Jr., and Hasan, I. (2006). Bank relationships and small firms' financial performance. Federal Reserve Bank of Atlanta. *Working paper 2006-5*.
- Child, J., and Terence, T. (2005). The dynamic between firms' environmental strategies and institutional constraints in emerging economies: Evidence from China and Taiwan. *Journal of Management Studies*, Vol. 42, 95-125.
- Cull R., Haber, S., and Imai, M. (2011). Related lending and banking development, and policy. *Research Working Paper*, No. 5570. The World Bank Development Research Group, Finance and Private Sector Development Team.
- Degryse, H., and Cayseele, P. V. (2000). Relationship lending within a bank-based system: Evidence from European small business data. *Journal of Financial Intermediation*, 9, 90-109.
- Degryse, H., and Ongena, S. (2001). Bank relationships and firm profitability. *Financial Management*, Spring, 9-34.
- Deloof, M., and Vermoesen, V. (2010). The value of bank relationships: Evidence from Belgium at the start of the Great Depression. *Working paper*. University of Antwerp and Louvain School of Management.
- Detragiache, E., Garella, P.G., and Guiso, L. (1999). Multiple versus single banking relationships: Theory and evidence. *Journal of Finance*, Vol. 55, No. 3 (June), 1133-1161.
- Dewatripont, M., and Maskin, E. (1995). Credit and efficiency in centralized and decentralized economies. *Review of Economic Studies*, 62(4), 541–55.
- Diamond, D., and Rajan, G. (1998). Liquidity risk, liquidity creation and financial fragility: A theory of banking, *Journal of Political Economy*, Vol. 109, no. 2.
- Diamond, D. (1984). Financial intermediation and delegated monitoring. *Review of Economic Studies*, *Vol. 51*, 393-414.
- Diamond, D. (1991). Monitoring and reputation: The choice between bank loans and privately placed debt. *Journal of Political Economy*, 99, 689-721.
- Elsas, R. (2005). Empirical determinants of relationship lending. *Journal of Financial Intermediation*, 14, 32-57.
- Ergungor, O. E., and Thomson, J. B. (2005). Industrial loan companies. *Economic Commentary*.
- Fama, E. (1985). What's different about banks? Journal of Monetary Economics, 15, 29-39.
- Fama, E. F., and Jensen, M. C. (1983). Separation of ownership and control. *Journal of Law and Economics*, 26(June), 301-325.

- Farinha, L. A., and Santos, A. C. (2002). Switching from single to multiple bank lending relationships: Determinants and implications, *Journal of Financial Intermediation*, Vol. 11, 124-151.
- Fazzari, S. M., Hubbard, R. G., and Peterson, B. C. (1988). Financing constraints and corporate investment. *Brookings Papers on Economic Activity*, 141-206.
- Foglia, A., Laviola, S., and Marullo, R, P. (1998). Multiple banking relationships and the fragility of corporate borrowers. *Journal of Banking and Finance*, 22, 1441-1456.
- Fok, C. W., Chang, Y. C., and Lee, W. T. (2004). Bank relationships and their effects on firm's performance around the Asian financial crises: Evidence from Taiwan. *Financial Management*, Summer, 89-112.
- Garriga, M. J. (2006). The effect of relationship lending on firm performance. Department of Business Economics, Universitat Autonoma de Barcelona.
- Geroski, P. A., and Mata, J. (2005). Founding conditions and the survival of new firms.
- Harhoff, D., and Korting, T. (1998). Lending relationships in Germany Empirical evidence from survey data. *Journal of Banking and Finance*, Vol. 22, 1317–1353.
- Haubrich, J. (1989). Financial intermediation, delegated monitoring, and long-term relationships, *Journal of Banking and Finance*, 13, 9-20.
- Heffernan S. (2007). Nowoczesna bankowość, PWN, Warsaw.
- Hellwig, M., (1991). Banking, financial intermediation and corporate finance, in Giovannini and C.Mayer, eds., *European financial integration* (Cambridge University Press, Cambridge) 35-63.
- Hiraki, T., Ito, A., and Kuroki, K. (2003). Single versus multiple main bank relationships: Evidence from Japan. *Social Science Research Network*.
- Hodgman, D. R. (1961). The deposit relationship and commercial bank investment behavior, *Review of Economics and Statistics*, 41, 257-261.
- Hodgman, D. R. (1963). Commercial bank loan and investment policy. *Bureau of Economic* and Business Research, University of Illinois, Urbana-Champaign.
- Holmstrom, B., and Tirole, J. (1997). Financial intermediation, loanable funds and the real sector. *Quarterly Journal of Economics*, 112, 663-69.
- Horiuchi, A., Packer, F. and Fukuda, S. (1988). What role has the "Main Bank" played in Japan? *Journal of the Japanese and International Economies*, 2, 159-180.
- Hoshi, T., Kashyap, A., and Scharfstein, D. (1991). Corporate structure, liquidity and investment: Evidence from Japanese industrial groups. *Quarterly Journal of Economics*, 106, 33-60.

Houston, J., and James, C. M. (1996). Bank information monopolies and the mix of private and public debt choices. *Journal of Finance*, 51, 1863-1889.

http://www.mpiewjena.mpg.de/english/news/workshops/egpworkshop/Geroski\_etal.pdf

- James, C. (1987). Some evidence on the uniqueness of bank loans. Journal of Financial Economics, 19, 217-235.
- Japelli, T., and Pagano, M. (2002). Information sharing, lending and defaults: Cross-country evidence. *Journal of Banking and Finance*, 26, 2017–2045.
- Jensen, M. C., and Meckling, W. (1976). The theory of the firm: Managerial behavior, agency costs, and ownership structure. *Journal of Financial Economics*, 3, 305-360.
- Kahn, C., and Winton, A. (1998). Ownership structure, speculation, and shareholder intervention. *Journal of Finance*, 53 (1), 99-129.
- Keeley, M. (1990). Deposit insurance, risk and market power in banking: American economic review, 80, 1183-200.
- Koch, T.W., and MacDonald, S.S. (2000). Bank management. 4th edition. USA: The Dryden Press.
- Kutsuna, K., Smith, K. J., and Smith, L. R. (2003). Banking relationships and access to equity capital markets: Evidence from Japan's main bank system. Claremont Colleges. *Working paper*.
- Leland, H. E., and Pyle, D. H (1977). Informational asymmetries, financial structure, and financial intermediation. *The Journal of Finance, Vol.* 32, 371-387.
- Levine, R. (2002). Bank-based or market-based financial systems: Which is better? *Journal* of Financial Intermediation, 11 (4), 398-428.
- Limpaphayom, P., and Polwitoon, S. (2004). Bank relationship and firm performance: Evidence from Thailand before the Asian financial crisis. *Journal of Business Finance and Accounting*, 31, 1577-1600.
- Longhofer, S. D. and Santos, J. A. C. (2000). The importance of bank seniority for relationship lending. *Journal of Financial Intermediation*, *Vol.* 9, 57–89.
- Maher, M., and Andersson, T. (1999). Corporate governance: Effects on firm performance and economic growth. Organisation for Economic Co-Operation and Development Report.
- Massa, M., and Zahid Rehman. (2008). Information flows within financial conglomerates: Evidence from the banks-mutual funds relationship. *Journal of Financial Economics*, 89 (2), 288-306.
- Mata, J., and Portugal, P. (1994). Life duration of new firms. *The Journal of Industrial Economics*, 42(3), 227-245.

- Morck, R., and Nakamura, L. (1999). Banks and corporate control in Japan. *The Journal of Finance. Vol. LIV*, No. 1, Feb., 319-339.
- Nakamura, L. (1993). Recent research in commercial banking: Information and lending. *Financial Markets Institutions and Instruments, Vol.* 2, 73-88.
- Nakamura, L. I. (1993). Commercial bank information: Implications for the structure of banking, in Klausner, Michael, and Lawrence J. White, eds.: *Structural Change in Banking*, 131-160.
- Ongena, S., and Smith, D. C. (2000). Bank relationship: A survey. *In Harker and Zenios eds*. The Performance of Financial Institutions, Cambridge University Press, 2000.
- Ongena, S., Smith, D. C., and Michalsen, D. (2003). Firms and their distressed banks: Lessons from the Norwegian Banking Crisis (1988-1991). *Journal of Financial Economics*, 67, 81112.
- Padilla, J. A., and Pagano, M. (1997). Endogenous communication among lenders and entrepreneurial incentives. *Review of Financial Studies*, 10, 205–36.
- Peltoniemi, M. (2004). Cluster, value network and business ecosystem: knowledge and innovation approach". A paper presented at the conference "Organisations, innovation and complexity: New perspectives on the knowledge economy", University of Manchester.
- Petersen M., and Rajan, R. (1994). The benefits of lending relationships: Evidence from small lending business data. *Journal of Finance*, Vol. 49, 3 37.
- Petersen, M., and Rajan, R. (1995). The effect of credit market competition on lending relationships. *Quarterly Journal of Economics*, 110, 406-443.
- Petrunia, R. (2002). Start-up conditions and the post-entry experience of new firms. http://www.sauder.ubc.ca/research/research\_centres/era/arena1/papers/paper%202.pdf
- Rajan, R. G. (1992). Insiders and outsiders: The choice between informed and arm's length debt. *Journal of Finance*, 47, 1367-1400.
- Sharpe, S. A. (1990). Asymmetric information, bank lending and implicit contracts: A stylized model of customer relationships. *Journal of Finance*, 45, 1069-1087.
- Shen, C. H., and Wang, C. A. (2004). Does bank relationship matter for a firm's investment and financial constraints? The case of Taiwan. *Pacific Basin Financial Journal*, 13, 163-184.
- Slovin, M. B., Sushka, vM. E., and Polonchek, J. A. (1993). The value of bank durability: Borrowers and bank stakeholders. *Journal of Finance*, 247-66.
- Stiglitz, J. E., and Weiss, A. (1981). Credit rationing with imperfect information. *The American Economic Review*, *Vol.* 71, 393-410.

- Strahan and Weston, (1998). Small business lending and the changing structure of the banking industry, *Journal of Banking and Finance*, 22, 821-845.
- Thakor, A. V. (1996). Capital requirement, monetary policy, and aggregate bank lending: Theory and empirical evidence. *The Journal of Finance, Vol.51*, 279-324.
- Thanh, Vu Huu, and Minh Ha, Nguyen. (2013). The effect of banking relationship on firm performance in Vietnam. *International Journal of Economics and Finance, Vol. 5*, No. 5; 148-158.
- Tian, L., and Estrin, S. (2008). Retained state shareholding in Chinese PLCs: Does government ownership always reduce corporate value? *Journal of Comparative Economics*, 36, 74-89.
- Tobin, J. (1969). A general equilibrium approach to monetary theory. *Journal of Money, Credit and Banking*, 1 (1), 15–29.
- Udell, G.F. (1989). Loan quality, commercial loan review and loan officer contracting. *Journal of Banking and Finance*, *Vol.13*, 367-382.
- Vale, B. (1993). The dual role of demand deposits under asymmetric information, Scandinavian, *Journal of Economics*, 95, 77-95.
- von Rheinbaben J., and Ruckes, M. (2004). The number and the closeness of bank relationships. *Journal of Banking and Finance, Vol.* 28, 1597-1615.
- von Rheinbaben, J., and Ruckes, M. (1998). A Firm's optimal number of bank relationships and the extent of information disclosure, University of Mannheim.
- von Thadden, E. L. (1995). Long term contracts, short term investment, and monitoring. *Review of Economic Studies*, 62, 557-575.
- von Thadden, E. L., (1992). The commitment of finance, duplicated monitoring and the investment horizon.
- von Thadden, E. L., (1998). Asymmetric information, bank lending, and implicit contracts: the winner's curse, University of Lausanne (DEEP).
- Wei, Z., Xie, F., and Zhang, S. (2005). Ownership structure and firm value in China's privatized firms: 1991-2001. *Journal of Financial and Quantitative Analysis*, 40 (1), 87-108.
- Weinstein, D., and Yafeh, Y. (1998). On the costs of a bank-centered financial system: Evidence from the changing main bank relations in Japan. *Journal of Finance*, 53, 635-672.
- Wood, J. (1995). Commercial Bank Loan and Investment Behavior. John Wiley and Sons, London.

- Yosha, O. (1995). Information disclosure costs and the choice of financing source. *Journal of Financial Intermediation*, 4, 3-20.
- Zahra, S. A., and Pearce, J. A. (1989). Boards of directors and corporate financial performance: A review and integrative model. *Journal of Management*, 291-334.